Author’s response to reviews

Title: Intrahousehold management and use of nutritional supplements during the hunger gap in Maradi region, Niger: a qualitative study

Authors:

Caroline Marquer (caroline.marquer@epicentre.msf.org)
Céline Langendorf (celine.langendorf@epicentre.msf.org)
Lynda Christelle Woi-Messe (lmesse@unicef.org)
Fatou Berthe (fatou.berthe@epicentre.msf.org)
Eric Alain Ategbo (eaategbo@unicef.org)
Santiago Rodas-Moya (rodas.santiago@gmail.com)
Saskia dePee (saskia.depee@wfp.org)
Rebecca F. Grais (rebecca.grais@epicentre.msf.org)

Version: 4 Date: 06 Dec 2019

Author’s response to reviews:

December 6, 2019
Editor-in-Chief
BMC Nutrition
Reference: Submission of a research article to be considered for publication in BMC Nutrition

Dear Editor,
We are writing to respectfully submit the revisions of our manuscript entitled “Intrahousehold perception and management of nutritional supplements during the hunger gap in Niger: a qualitative study.” We have revised the manuscript according to the editor’s comments and responses are provided below.
We appreciate your consideration of our manuscript and look forward to hearing from you.
Sincerely,
Caroline Marquer
Epicentre
Tel: +33 1 40 21 54 63
E-mail: caroline.marquer@epicentre.msf.org

Responses to the comments
1. Overlap
The section on data collection and analysis has been revised accordingly and does not contain any overlap. We thank the editor for highlighting this issue.
2. Role of Funder
The role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript was added.

3. Authors’ Contributions
The section was revised accordingly.

4. Interviews
The interview guide was developed specifically for this study. We added the interview guide and included it as an additional file. We also cite it in the main manuscript.

5. List of Abbreviations
The sections was renamed accordingly.

6. Clean manuscript
The manuscript was revised accordingly, and a clean version provided.